Author's response to reviews

Title: Ovarian dysfunction and FMR1 alleles in a large Italian family with POF and FRAXA disorders: Case report

Authors:

Maria Giuseppina Miano (miano@igb.cnr.it)
Carmela Laperuta (laperuta@igb.cnr.it)
Pietro Chiurazzi (pietro.chiurazzi@rm.unicatt.it)
Michele D'Urso (durso@igb.cnr.it)
Matilde Valeria Ursini (ursini@igb.cnr.it)

Version: 4 Date: 5 April 2007

Author's response to reviews:

Institute of Genetics and Biophysics "Adriano Buzzati Traverso"
Via P. Castellino, 111-80131, Napoli, Italy

Naples 05.04.2007

Dear Prof. Liz Hoffman
Assistant Editor
BMC-series journals

Dear Editor,
I made the following changes to the format the paper Miano et al, "Ovarian dysfunction and FMR1 alleles in a large Italian family with POF and FRAXA disorders: Case report" as requested. Herein, I reported the point-by-point responses.

Sincerely,
Maria G Miano

Formatting changes requested
---------------------------------------------
1. Underlining - Please do not underline any section headings.
2. Running title - Please remove the running title from the title page.
3. Keywords section - Please remove the keywords section from the manuscript.
4. Manuscript sections - Manuscript sections should include (in the following order): Abstract; Background; Case Presentation; Conclusions.
5. Please move the figure legend to after the References.
6. Figures - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure cropping guide may be of some help, and can be viewed here: http://www.biomedcentral.com/info/ifora/cropping
   I minimized the white space around the image in the Figure that in the new version was cropped.